Author’s response to reviews

Title: Nine Veno-Venous ECMO Challenges and Controversies

Authors:

David B Tulman (dbtulman@gmail.com)
Saarik C Gupta (saarik.gupta@gmail.com)
Ravi S Tripathi (ravi.tripathi@osumc.edu)
Bryan A Whitson (bryan.whitson@osumc.edu)
Michael S Firstenberg (msfirst@gmail.com)
Don Hayes (don.hayes@nationwidechildrens.org)
Xuzhong Xu (xuzhong@263.net)
Thomas J Papadimos (thomas.papadimos@osumc.edu)

Version: 2 Date: 11 September 2013

Author’s response to reviews:

Editors:

The bottom of page 17 has the permissions indications. One patient has a consent signed over to the hospital because she is our poster child for ECMO and does PR for us. The other patient is in a nursing home and their permission is in our EPIC electronic medical record. If needed we will provide them. Dr. Bryan Whitson (one of the authors) was their surgeon.